Author’s response to reviews

Title: Ethnic Differences in Ischemic Stroke Subtypes in Young-Onset Stroke: The Stroke Prevention in Young Adults Study

Authors:

Megh M Trivedi (megh.trivedi@som.umaryland.edu)
Kathleen A Ryan (kryan@medicine.umaryland.edu)
John W Cole (jcole@som.umaryland.edu)

Version: 3  Date: 22 February 2015

Author’s response to reviews:

Hello,

Per the request of the editorial office (Jhonell De Los Santos) the following statements were added to the end of the first paragraph of the Methods section:

A written informed consent for participation in the study was obtained from each participant or when appropriate from the participant’s guardian. This protocol was approved by University of Maryland, Baltimore Institutional Review Board prior to study commencement.

Thank you, J Cole